Author's response to reviews

Title: The use of a lumbar drain for management of pseudotumor cerebri caused by All-trans retinoic acid

Authors:

Fahid T Rasul (fahidrasul@doctors.org.uk)
Ahmed K Toma (ahmedktoma@yahoo.com)
Akbar A Khan (akbaralikhan77@hotmail.com)
Gordon T Plant (gordon.plant@uclh.nhs.uk)
Laurence D Watkins (laurence.watkins@uclh.nhs.uk)

Version: 5 Date: 4 October 2012

Author's response to reviews: see over
To Whom It May Concern:

Re: The use of a lumbar drain for management of pseudotumor cerebri caused by All-trans retinoic acid

I attach a revised manuscript following editorial review and comments. Here I outline the revisions made:

Editors comments:

In their revised manuscript the authors mentioned that they had not done cerebral venogram. They mentioned even if they had found sinus thrombosis on the venogram, there was not much that they could/would have done. I think if they had found venous sinus thrombosis, they might have contemplated or instituted anti-coagulation therapy with heparin infusion or therapeutic clexane.

Our response:

We agree that a CT venogram may be a useful diagnostic tool for cerebral venous sinus thrombosis. We have suggested this in the manuscript. In addition we have altered the manuscript to suggest that it may lead to an alternative treatment in the form of anticoagulation therapy if found to be positive. We thank the editor for their comments.

The revised manuscript shows the alterations made in ‘tracked changes’ form.

Thank you for once again for considering publishing our article in your journal. We look forward to hearing from you.

With best wishes
Yours sincerely
Mr Fahid Rasul (MRCS, MBBS, BSc)
Victor Horsley Department of Neurosurgery
National Hospital for Neurology and Neurosurgery
Queen Square
London, UK.
(on behalf of all the authors)